Author's response to reviews

Title: BAFF promoted proliferation of human mesangial cells through interaction with BAFF-R

Authors:

- Nuoyan Zheng (zhnuoy@mail.sysu.edu.cn)
- Dongxian Wang (2592272098@qq.com)
- Hongyan Ming (mhy_lucky@163.com)
- Haiqing Zhang (qingzh123@163.com)
- Xueqing Yu (yuxq@mail.sysu.edu.cn)

Version: 2
Date: 19 November 2014

Author's response to reviews:

Dear editor,

Here we present our study "BAFF promoted proliferation of human mesangial cells through interaction with BAFF-R" for your consideration in "BMC nephrology" journal. According to your request, the ethics statement has been included in "Material and Methods" in revised new manuscript. Also, consents of participant are also stated in "Material and Methods".

We sincerely hope that you can consider our work for your journal and your comments are very welcomed.

With Warm Regards

Xueqing Yu MD, PhD
Professor of Medicine
Director, Institute of Nephrology
Vice President, The First Affiliated Hospital
Sun Yat-Sen University